Author's response to reviews

Title: The predictors of glucose screening: the contribution of risk perception
Lavielle P 1,2,* and Wacher N 1 1 Clinical Epidemiology Research Unit. Hospital de Especialidades, Centro Medico Nacional Siglo XXI. Instituto Mexicano del Seguro Social. 2 Hospital General de Mexico. Federal Health Ministry, Mexico City, Mexico. *Corresponding author: Lavielle Pilar. Hospital de Especialidades, Centro Medico Nacional Siglo XXI. Instituto Mexicano del Seguro Social Cuauhtemoc 330, Colonia Doctores, 06720, Cuauhtemoc, Distrito Federal. Mexico City, Mexico. Phone: 5255-5627-6900 ext. 21481 and 21507. Fax: 5255-1054-1599

Authors:

PILAR LAVIELLE (lavielle.pilar@gmail.com)
NIELS WACHER (nwacher@hotmail.com)

Version: 2 Date: 18 October 2013

Author's response to reviews: see over
Please find enclosed our manuscript: The predictors of glucose screening: the contribution of risk perception, which we are sending to BMC Family Practice for your editorial consideration.

The prevention of type 2 diabetes is a challenge for health institutions. Risk perception is an important construct to study, due to it has implications for future health policies, specifically in risk communication activities that may lead people to adopt preventive behaviors, and identifying predictors of specific behavior such as glucose screening could be beneficial to those with a potential for contracting type 2 diabetes.

This study has neither published nor simultaneously submitted to any other journal. All authors have participated in writing and approval of this submission, and each can take public responsibility for the content.

We made some changes such as include a title at top of the manuscript and the modifying the original abstract.

I hope you will find this work suitable for publication at your prestigious journal.

Truly yours,

Dra. Pilar Lavielle Sotomayor